Author’s response to reviews

Title: A Rare Case of Pericarditis and Pleural Empyema Secondary to Transdiaphragmatic Extension of Pyogenic Liver abscess

Authors:

Eunae Cho (cea202@hanmail.net)
Sang Woo Park (withpsw@naver.com)
Chung Hwan Jun (estevanj@naver.com)
Sang Soo Shin (kjradsse@jnu.ac.kr)
Eun Kyu Park (iameunkyu@gmail.com)
Kyo Seon Lee (waytogosun@naver.com)
Seon Young Park (drpsy@naver.com)
Chang Hwan Park (p1052ccy@hanmail.net)
Hyun Soo Kim (dshskim@jnu.ac.kr)
Sung Kyu Choi (choisk@jnu.ac.kr)
Jong Sun Rew (jsrew@jnu.ac.kr)

Version: 3 Date: 03 Jan 2018

Author’s response to reviews:

Point-by-point response letter

1. Please confirm that the informed consent for publication obtained from the patient gave was in the written format. Please state this in the ‘Consent for publication’ section.

=> Thank you for your advice. We add the paragraph written below. (Consent for publication section, page 11)

“Written informed consent was obtained from the patient for publication of this case report.”
2. We would also like to ask for you to provide more justification for the contributions of SYP, CHP, HSK, SKC and JSR, as currently they do not automatically qualify for authorship. Contribution to reviewing the article, alone, does not usually justify authorship.

(=> Thank you for your advice. We add the paragraph written below. (Authors’ contributions section, page 12))

“EC, SWP designed the study and wrote the manuscript. CHJ participated in the design of the study and the study’s coordination. SSS contributed to the imaging and interpretation. EKP and KSL performed surgical treatment. SYP managed the patient and helped to draft the manuscript. CHP, HSK acquired and analyzed the laboratory and clinical data. SKC, JSR participated in reviewing the literature and editing the initial manuscript. All authors read and approved the final manuscript”

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

(=> Thank you for your advice. We revised the manuscript files as you recommended.)